BMJ Open Public reporting of clinical trial findings as an ethical responsibility to participants: a qualitative study

Richard L Morrow ,¹ Barbara Mintzes ,² Garry Gray,³ Michael R Law,⁴ Scott Garrison ,⁵ Colin R Dormuth⁶

To cite: Morrow RL, Mintzes B, Gray G, *et al.* Public reporting of clinical trial findings as an ethical responsibility to participants: a qualitative study. *BMJ Open* 2023;**13**:e068221. doi:10.1136/ bmjopen-2022-068221

Prepublication history and additional supplemental material for this paper are available online. To view these files, please visit the journal online (http://dx.doi.org/10.1136/ bmjopen-2022-068221).

Received 10 September 2022 Accepted 08 March 2023



© Author(s) (or their employer(s)) 2023. Re-use permitted under CC BY-NC. No commercial re-use. See rights and permissions. Published by BMJ.

For numbered affiliations see end of article.

Correspondence to

Dr Richard L Morrow; richard.morrow@ubc.ca

ABSTRACT

Objective To understand how the experiences and views of trial participants, trial investigators and others connected to clinical trial research relate to whether researchers have a duty to participants to publicly report research findings.

Design Qualitative interview study.

Setting Semistructured interviews held in person or by telephone between March 2019 and April 2021 with participants in the Canadian provinces of Alberta, British Columbia and Ontario.

Participants 34 participants, including 10 clinical trial participants, 17 clinical trial investigators, 1 clinical research coordinator, 3 research administrators and 3 research ethics board members.

Analysis We conducted a thematic analysis, including qualitative coding of interview transcripts and identification of key themes.

Main outcome measures Key themes identified through qualitative coding of interview data.

Results Most clinical trial participants felt that reporting clinical trial results is important. Accounts of trial participants suggest their contributions are part of a reciprocal relationship involving the expectation that research will advance medical knowledge. Similarly, comments from trial investigators suggest that reporting trial results is part of reciprocity with trial participants and is a necessary part of honouring informed consent. Accounts of trial investigators suggest that when drug trials are not reported, this may undermine informed consent in subsequent trials by withholding information on harms or efficacy relevant to informed decisions on whether to conduct or enroll in future trials of similar drugs.

Conclusion The views of trial participants, trial investigators and others connected to clinical trial research in Canada suggest that researchers have an obligation to participants to publicly report clinical trial results and that reporting results is necessary for honouring informed consent.

INTRODUCTION

A systematic review indicated approximately 4 in every 10 randomised controlled trials included in trial registries were not reported in journal articles after a period of 2 or more years from study completion.¹ Clinical studies

STRENGTHS AND LIMITATIONS OF THIS STUDY

- ⇒ The use of qualitative interviews allowed for an indepth exploration of experiences and views relating to clinical trial reporting as an ethical responsibility towards trial participants.
- ⇒ The study was strengthened by the inclusion of a range of participants, including past trial participants, trial investigators, research administrators and research ethics board members.
- ⇒ As the sample of past trial participants interviewed for this study was small, caution is warranted in generalising from these interviews.
- ⇒ The study focused on clinical trials in Canada, so it is not clear to what extent our findings apply to clinical trials in other countries.
- ⇒ It is possible that attitudes towards clinical trial reporting differed in those who participated compared with those who did not take part in the study.

with results favourable for the experimental treatment are more likely to be published, leading to bias in the medical literature.² Studies have also found low compliance with regulatory requirements for timely reporting of clinical trial results within ClinicalTrials. gov and the EU Clinical Trials Register.^{3 4} A study of clinical trials conducted in Canada, including trials registered in ClinicalTrials. gov and completed between 2009 and 2019, found that only 39% of trials had reported results in the registry by early October 2021.⁵ Moreover, the problem of non-publication of clinical trials has been documented across many areas of medicine⁶⁻¹⁰ and, although more frequent in earlier phase trials, for all phases of clinical trials.² Selective reporting of trials has led to less informed patient care, unnecessary harm to patients and a waste of research resources.^{11-f:}

Advocates of full reporting of clinical trials have argued non-publication betrays trial participants and violates an implicit contract between participants and researchers.^{14–18} They reason that when individuals agree

Open access

to participate in trials, they expect their participation will contribute to medical knowledge and help future patients. When trial findings are not reported, this expectation is not fulfilled. More fundamentally, as individuals may reasonably expect trials to contribute to knowledge when deciding to participate in a trial, non-reporting of clinical trials may undermine informed consent.^{17 18}

Arguments that clinical trial investigators have a duty to trial participants requiring them to report findings are strengthened by previous research suggesting that motivations for participation in trials include altruism.^{19–22} In addition, a survey of non-critically ill patients in an emergency department setting found that most felt it was important to make clinical trial results publicly available.²³ However, trial participant views on the importance of reporting research findings and trial investigator views on the responsibility to report findings are unclear.

We conducted a qualitative study of clinical trial reporting in Canada. Our broader study aimed to investigate factors contributing to non-publication and publication bias in clinical trials and related ethical issues.^{24 25} The analysis reported in this paper aimed to understand how the experiences and views of trial participants, trial investigators and others relate to whether researchers have a duty to trial participants to publicly report research findings in journals or trial registries.

METHODS

Study design

We conducted a qualitative study involving semistructured interviews and thematic analysis of interview data to investigate experiences and views related to clinical trial reporting.²⁶ This study used a qualitative approach because this provided the flexibility to investigate emergent themes and to collect rich data regarding views and experiences relating to reporting of clinical trials—for example, through prompting interviewees to elaborate on unanticipated or important points raised in their responses.^{27 28} The research team for this project included a clinical trial investigator (SG), an expert in qualitative methods (GG), a health research analyst (RLM) and researchers in epidemiology and health policy (BM, MRL and CRD). The researchers had no prior relationship with those interviewed for the study.

Sampling

We sought to interview clinical trial participants, clinical trial investigators, clinical research coordinators, research administrators and research ethics board (REB) members (inclusion criteria are listed in table 1). We used a purposive sampling strategy to involve trial participants who had participated in trials for a range of treatments, trial investigators in diverse medical fields and others connected to clinical trial research to provide additional perspectives. Snowball sampling was used to as a complementary strategy to gain referrals to additional trial investigators and REB members. We chose to include

Table 1 Types of interviewees and inclusion criteria				
Interviewee type	Inclusion criteria	Rationale		
Past trial participant	Participated in ≥1 clinical drug trial while at least 18 years of age; participation in the 5 years prior to interview, but has now ended	Will have experience related to trial participation and trial reporting		
Clinical trial investigator	Conducted ≥1 clinical drug trial	Will have experience relevant to trial reporting		
Clinical research coordinator	Coordinated ≥1 clinical drug trial	May have experience relevant to clinical trial reporting		
Research administrator	Knowledge of policy and practice related to dissemination of clinical trial findings and/or relations with trial sponsors	experience, knowledge and views from policy		
Clinical REB member	≥1 year of experience as clinical REB member	Experience in ethical review and familiarity with practice and policy relating to clinical trial reporting		

Created by the authors. A version of this table has previously been published, and it is reprinted in a modified form with permission.²⁴ REB, research ethics board.

in our study sample both those who had participated in trials and those involved in the conduct, administration or ethical review of trials for two reasons: first, each type of interviewee might provide insights from a different perspective, and second, this would allow for triangulation of findings.²⁷

Recruitment

Strategies to recruit past trial participants included advertising in a free newspaper and requesting assistance from clinical research coordinators and managers, who sought consent from past trial participants for us to contact them about our study. We emailed or telephoned 11 individuals who expressed interest following the advertisement or consented to be contacted (10 participated and 1 did not respond). We identified other types of interviewees through online sources (ClinicalTrials.gov, Canadian Clinical Trials Asset Map database and websites of research institutions and REBs) and referrals. We invited participation from 61 trial investigators by email (17 investigators participated, 2 responded but were unavailable for an interview during the study, 36 did not respond and 6 declined). Investigators who declined stated they were too busy (n=1), not interested (n=1) or lacked relevant experience (n=4). A clinical research coordinator

who worked with a participating trial investigator also volunteered to take part in an interview. In addition, we emailed 12 research administrators (3 participated and 9 did not respond) and 15 REB members (3 participated and 12 did not respond). Trial participants and trial investigators were eligible to receive a US\$50 honorarium for participation.

Data collection

We conducted semistructured interviews from March 2019 to April 2021. Interview guides for each type of interviewee were used (online supplemental appendix). Interviews were primarily based on open-ended questions and allowed for exploration of unanticipated issues. Data collection included initial interviews in person or by telephone with 34 individuals and follow-up telephone interviews with 4 individuals to collect additional information. The duration of interviews was approximately 45-60 min for initial interviews and 20 min for follow-up interviews. In-person interviews were held in a public library meeting room or at the interviewee's workplace. RLM conducted the research interviews and coded the interview data. Data collection continued until the data allowed for a detailed analysis addressing the study's research questions.

Data analysis

Interviews were audiorecorded and transcripts were analvsed using ATLAS.ti (V.8), including coding and deriving themes from the data. Analysis included initial coding with an open-ended approach, followed by additional analysis to retain and develop key themes for analysis.²⁶ Collection of data from different types of interviewees allowed for triangulation of data during analysis.²⁷ The Consolidated Criteria for Reporting Qualitative Studies checklist was used to guide reporting of findings.

Patient and public involvement

A patient advocate was consulted during the planning of this study regarding the importance of pursuing this research and strategies for recruiting past trial participants for interviews. All participants in this study who are interested will receive a summary of the study results, including past trial participants who took part in a research interview.

RESULTS

Overall, 34 individuals took part in the study, including 10 clinical trial participants, 17 clinical trial investigators, 1 clinical research coordinator, 3 research administrators and 3 REB members. (See table 2 for interviewee characteristics). The study included individuals from the Canadian provinces of Alberta, British Columbia and Ontario. Past trial participants varied by sex (three male; seven female), age (38–77 years at the time of their initial interview) and highest level of education completed (from elementary to university). They had taken part in trials of 6 months to 5 years in duration, testing treatments

Table 2 Interviewee characteristics					
Characteristic	Past trial participants (n=10)				
(A) Past trial participants					
Sex					
Female	7				
Male	3				
Age					
≤65 years	5				
>65 years	5				
Education, highest level completed					
Elementary	1				
Secondary	3				
Community college	1				
University	5				
Province					
Alberta	3				
British Columbia	7				
(B) Trial investigators					
Characteristic*	Trial investigators (n=17)				
Primary appointment					
University or academic teaching hospital	10				
Other (eg, private practice, cancer centre)	7				
Experience in role					
≤5 years	0				
>5 years	17				
Province					
Alberta	0				
British Columbia	9				
Ontario	8				
Types of funding					
Non-industry only	0				
Industry only	0				
Both industry and non-industry	17				
Most senior role					
Principal investigator for site	3				
Principal investigator for trial	14				
Trial type					
Single site only	0				
Multiple site only	1				
Both single and multiple site	16				
(C) Clinical research coordinators, research administrators and research ethics board (REB) members					

ethics board (REB) members

Characteristic	Research coordinator (n=1)	Research administrators (n=3)	REB members (n=3)
Primary appointment			
University or academic teaching hospital	0	3	1
Other (eg, private practice, cancer centre)	1	0	2

Continued

Table 2 Continued

(C) Clinical research coordinators, research administrators and research ethics board (REB) members

Characteristic	Research coordinator (n=1)	Research administrators (n=3)	REB members (n=3)
Experience in role			
≤5 years	0	0	1
>5 years	1	3	2
Province			
Alberta	0	1	1
British Columbia	1	0	2
Ontario	0	2	0

Table 2 was created by the authors. A version of table 2 has previously been published, and it is reprinted in a modified form with permission.²⁴

*Classifications were based on those used for an investigator survey by Rochon *et al.*³⁸ BEB research ethics board

for cardiovascular disease, *Clostridioides difficile* infections, chronic pain, diabetes, eye disorders and multiple sclerosis. Among interviewees who were involved in the conduct, administration or ethical review of trials, some spoke about both conducting trials and playing a role in research administration or reviewing ethics applications. Trial investigators who took part in the study had conducted trials in cardiovascular medicine, endocrinology, hepatology, infectious diseases, oncology, psychiatry and rheumatology.

Our study results are presented below by theme. This includes themes relating to trial participant experiences and views (motivations for participating in a trial and trial participant views on reporting research findings), accompanied by quotations from trial participants (P1–P10). The findings below also include themes related to accounts from those involved in the conduct, administration or ethical review of trials (views on clinical trial reporting as a responsibility to research participants and linking clinical trial reporting to informed consent), presented with quotations from trial investigators (T1–T17).

Views of clinical trial participants

Motivations for participating in a clinical trial

Most trial participants stated they were motivated to take part in the clinical trial in part to help future patients. Patients with a less urgent need to improve their health condition were more likely to identify helping future patients as their primary reason for joining a trial. For example, a patient with type 1 diabetes recalled that she had joined multiple clinical trials over time because she 'figured if there isn't research being done to help people, then nothing is ever going to improve.' (P7) Patients who were in more urgent need to improve their health condition were more likely to identify access to treatment as their primary reason for participating in a trial. However, they also often wanted to help future patients and conceived of their participation as an act of solidarity with others like them. A patient who had experienced a recurring C. difficile infection recalled that at the time of joining a clinical trial she had told family members: 'Nobody should have to suffer this way, and if there's anything that I can do to help medical science move forward so that other people don't have to suffer like this in the future, I'm all for it.' (P6) Although most participants were motivated to join a trial in part to help others, this was not always the case. One trial participant, who joined a trial to access treatment after suffering from a sudden deterioration of her vision due to an eye disorder, recalled that her initial motivation was solely to address her own health needs. Trial participants typically had multiple reasons for participating in a trial. Motivations varied among participants but included access to free medication or medical supplies, access to better care and helping one's health provider or researchers.

Trial participant views on reporting research findings

Most past trial participants felt it was important for the results of clinical trials to be published. Trial participants stated various reasons they felt publishing research findings was important. Some suggested if results were not published, this would represent a waste of time or resources. A patient who had participated in a trial to test a treatment for C. difficile felt it was important to avoid wasting the effort and resources invested in a trial: 'If we're doing the work, spending the dollars and not using that information to further medical science, then what was the point of doing all that work in the first place?' (P6) Another patient, who had taken part in a trial to test a treatment for relapsing-remitting multiple sclerosis, emphasised the importance of reporting results to help future patients: 'If you don't publish... then how is it to be paid forward to help other people?' (P5) Some trial participants stated it was important to publish trials to learn from negative or incomplete trials, inform the medical community, demonstrate transparency and improve future research. However, not all patients stated that publishing trial results was important. One patient felt it was hard for her to judge whether it was important to publish results from a trial suggesting a treatment did not work. Another patient spoke about how he would feel about publication of results of the trial he participated in, rather than about the importance of reporting in general, saying: 'I would feel a little better knowing that my participation helped in something.' (P9)

When some participants described the value of clinical trial reporting, they highlighted the contribution of trial participants to research. A patient with type 1 diabetes felt it was important to publish trial results, 'because of a lot of effort that a lot of people put into it—not just the researchers, but the people that were participating in the trial.' (P7) Similarly, another trial participant said she felt it was important to publish trial results in part because 'people were gracious enough to be part of it.' (P2) One trial participant reflected that she was quite willing to be a 'guinea pig', but she would feel 'cheated' if the trial she had participated in were not published, because she

had participated 'not just for me.' (P5) Taken together with statements from a larger number of trial participants that they had participated in part to help other patients, these comments suggest reporting the results of trials is important as a form of reciprocity between researchers and trial participants. However, none of the trial participants was aware of whether the results of the trial they had participated in had been published, although in some cases the trials they had taken part in were either ongoing or so recent that it would be reasonable that results might not have been published at the time of their interview. In effect, reciprocity between trial participants and researchers may require reporting of trial results, but trial participants may often not be able to observe whether this is fulfilled.

Views of those involved in the conduct, administration or ethical review of trials

Clinical trial reporting as a responsibility to research participants

Among investigators, administrators and REB members interviewed for this study, many felt researchers have an obligation to trial participants to report the results of clinical trials. Comments highlighted that trial participants contribute their time and expose themselves to risk, yet may not directly benefit through their participation. Several comments suggested reporting results is necessary as a kind of reciprocity, or to fulfil an implicit agreement, between trial participants and researchers. A trial investigator who studied treatments for infectious diseases felt publishing was important as a responsibility to trial participants: 'Well, they've spent their time ... There's a potential risk of entering a clinical trial, so I think as researchers we have a responsibility to hold true to their commitment and altruism to enter into a clinical trial.' (T16) An endocrinologist who conducted clinical trials said: 'I think most people understand that this may or may not benefit them, but hopefully this will benefit society... If it's not even published, then we're not fulfilling our side of the bargain.' (T14) Similarly, a couple of investigators suggested that a 'contract' between participants and researchers obligated researchers to report results. Notably, the chronology of this reciprocity or 'bargain' involves the trial participants contributing their time and exposing themselves to risk without knowing whether researchers will fulfil their implicit obligation to report the research findings. This was reflected in the comment of one trial investigator, who noted: 'People have volunteered, given their time, given their samples in good faith that some science is going to come out of it.' (T7, emphasis added)

Some trial investigators felt a responsibility to trial participants to publicly report trials results existed but it could be difficult or less important to publish in certain circumstances. A cardiovascular investigator, who spoke about the difficulty of publishing incomplete trials, stated: 'To me, not publishing is unethical, but I can see some situations where it's just not possible.' (T10) He also suggested trials stopped early following a decision by an industry sponsor to halt development of a drug could be less important to publish due to a lack of statistical power and lack of relevance. Similarly, some interviewees mentioned that trials which are unable to recruit many patients may not be published and that it can be difficult to make inferences from or publish an incomplete trial. In addition, some investigators highlighted that a lack of time or resources may be a factor contributing to nonreporting or delays in reporting. While trial results could be reported in trial registries such as ClinicalTrials.gov, which could be particularly helpful if results were difficult to publish in a journal, not all trial investigators were aware of the possibility of reporting results in registries.

Linking clinical trial reporting to informed consent

Several investigators linked an obligation to report trial results in a journal or trial registry to informed consent. In some cases, consent forms signed by trial participants actually indicate research findings will be published. More generally, trial participants may reasonably expect or be told a trial will contribute to medical knowledge. An investigator in hepatology trials suggested this requires researchers to report their findings: 'We specifically say the benefit will be greater knowledge to the scientific and medical community, which will hopefully benefit other people in the future. So if we're not sharing the results of the study, then that's not true... We are not honouring that consent.' (T15)

When trials showing drug harms or a lack of efficacy, including early phase trials, are not reported, this may also undermine informed consent in future trials. An investigator noted that trials identifying safety concerns may provide information relevant to future trials of similar drugs. Although another investigator was less concerned about this issue because drugs in the same class would not necessarily be associated with the same adverse effect, in some cases information about harms of one drug in a class is deemed important enough to add to consent forms used in trials of drugs in the same class. Similarly, a trial investigator and an REB member involved in conducting trials each highlighted that publishing trials showing harms may inform other trialists that trials of the same or similar drugs would expose patients to excessive risk. In addition, the REB member commented that non-publication of negative trials may lead to redundant research which unknowingly involves patients in trials of drugs lacking efficacy.

DISCUSSION

The accounts of trial participants, trial investigators and others connected to clinical trial research suggest that when researchers enrol patients in clinical trials there is often an implicit understanding among researchers and trial participants involving an obligation to publicly report research results. Most trial participants were motivated to enter clinical trials in part to advance science, and most felt that reporting the results of clinical trials is important. Trial participant accounts suggest their contributions are part of a reciprocal relationship involving the expectation that research will advance medical knowledge. Similarly, comments from trial investigators suggest that reporting trial results is part of reciprocity with trial participants and is a necessary part of honouring informed consent. In addition, when trials are not reported, this may undermine informed consent in subsequent trials by withholding information on harms or efficacy relevant to informed decisions on whether to conduct or enrol in future trials of similar drugs.

Comparison with existing literature

Our finding that many trial participants were motivated to join trials in part to help future patients is consistent with previous studies on reasons for participation in trial research.^{19–22} Our study adds that even patients who are strongly motivated to participate by the opportunity to access treatment may feel it is important to help future patients out of a sense of solidarity with others like them.

A survey of non-critically ill patients in an academic emergency department in the northeastern USA found that most felt it was important to report trials results.²³ Our study indicated most individuals who had recently taken part in a clinical trial felt it was important to report trial results, while highlighting trial participants may view their own contributions as part of a reciprocal relationship involving the expectation that trials will contribute to medical knowledge. However, this reciprocity which involves a responsibility for researchers to report trial results may be weakened for various reasons. First, trial participants may often not find out whether trial results are published by the researchers, which might diminish a researcher's sense of the obligation to publish as a responsibility to the trial participants. Second, trial participants might be unlikely to question whether results have been reported, due to losing contact with researchers who are not their regular health providers, respect for the authority and expertise of the researchers, or gratitude for other benefits received in the trial (such as access to treatment or greater medical attention).

Importantly, our study strengthens empirical support for arguments that when trial results are not reported, this violates an implicit agreement or contract between researchers and participants and undermines informed consent.^{15–18} Trial participants may consent to enter a trial with the understanding that research will benefit future patients. However, this consent is not respected when trial results are not reported and this potential benefit is not fulfilled. In effect, the core ethical principle of respect for persons is undermined, as informing trial participants of the risks and benefits of research is part of respecting their autonomy as research participants.^{29 30}

Notwithstanding the responsibility to publicly report trial findings, some investigators noted it may be difficult to complete a trial due to low recruitment or other reasons, and it may be difficult to publish certain findings such as those from an incomplete trial. Estimates of the rate of discontinuation of clinical trials vary, but discontinued trials are less likely to be published.^{31–33} A focus group study of biomedical researchers found that many felt it was difficult to publish negative or 'no difference' results.³⁴ It may even appear questionable to report a trial that encountered problems such as difficulty in recruitment or high drop-out rates. However, trial registries provide an avenue to report trials that might be difficult to publish in a journal, and may, like ClinicalTrials.gov, allow authors to provide reasons for early termination of a trial and to describe limitations or caveats regarding a trial's results.³⁵

Policy implications

While the analysis reported in this paper highlights an ethical responsibility to report research results, this responsibility does not lie with trial investigators alone. Previously reported findings from our broader study of clinical trial reporting, based on the same interviews analysed for this paper, indicated that industry sponsors of clinical trials may exert influence on whether results are reported and that clinical trial reporting practices are shaped by incentives within the research system favouring publication of positive or negative trials, such funding opportunities and academic promotion, bonuses and recognition.²⁴²⁵ Similarly, a survey study found researchers were 'aware of being the main culprits of non-publishing or selective publishing of results from clinical trials' but felt that 'blame rested not solely with them but with the system that encourages and supports practices that lead to publication bias-from funders and research institutions to journals and trial registries.³⁶ The responsibility to ensure trials are publicly reported is shared by trial investigators with other actors in the research system who shape the context in which trial reporting takes place.

Stronger regulatory measures could improve clinical trial reporting policy or practices of research institutions, sponsors and individual investigators. In Canada, it is important to adopt regulatory measures to require reporting of clinical trial results within a recognised trial registry. While phase 1 trials are largely excluded from current regulatory reporting requirements in the USA and European Union,^{4 37} our study highlights reporting early phase trials is necessary for fulfilling informed consent. The effectiveness of mandatory reporting requirements and expanding their scope to cover all clinical trials of drugs and biologics.^{3 4 38}

Strengths and limitations of this study

The use of qualitative interviews allowed for an in-depth exploration of experiences and views relating to clinical trial reporting as an ethical responsibility towards trial participants. A strength of our study was the inclusion of a range of participants, including past trial participants, trial investigators, research administrators and REB members. Our study also had limitations. As the sample of past trial participants interviewed for this study was small, caution is warranted in generalising from these interviews. However, this limitation was mitigated by triangulation of findings among different types of interviewees regarding reciprocity between researchers and

triangulation of findings among different types of interviewees regarding reciprocity between researchers and trial participants and the responsibility to report results. More generally, it is not clear to what extent our findings apply to clinical trial settings outside Canada, as experiences and views of clinical trial reporting might vary due to differences in funding, policy or healthcare systems. In addition, it is possible that attitudes towards clinical trial reporting differed in those who participated compared with those who did not take part in the study.

CONCLUSIONS

The views of trial participants, trial investigators and others connected to clinical trial research in Canada suggest that researchers have an obligation to research participants to report clinical trial results and that reporting of results in registries or journals is necessary for honouring informed consent. This highlights the need for Health Canada to adopt regulatory measures to require timely reporting of clinical trial results within a recognised trial registry. Future studies could investigate views on clinical trial reporting in other countries.

Author affiliations

¹School of Population and Public Health, The University of British Columbia, Vancouver, British Columbia, Canada

²Faculty of Pharmacy and Charles Perkins Centre, University of Sydney Faculty of Health Sciences, The University of Sydney, New South Wales, Australia ³Department of Sociology, University of Victoria, Victoria, British Columbia, Canada ⁴Centre for Health Services and Policy Research, School of Population and Public Health, The University of British Columbia, Vancouver, British Columbia, Canada ⁵Family Medicine, University of Alberta, Edmonton, Alberta, Canada ⁶Anaesthesiology, Pharmacology, and Therapeutics, Therapeutics Initiative, The University of British Columbia, Vancouver, British Columbia, Canada

Twitter Richard L Morrow @InsideTrials

Acknowledgements We thank all the people who took time to participate in the study.

Contributors Guarantor: RLM is the guarantor and accepts full responsibility for the finished work and the conduct of the study, had access to the data, and controlled the decision to publish. Conception of study: RLM. Substantial contributions to design of the work: RLM, BM, GG, MRL and CRD. Acquisition and analysis of data: RLM. Interpretation of data: RLM, BM, GG, MRL, SG and CRD. Drafting the work: RLM. Revising the work critically for important intellectual content: RLM, BM, GG, MRL, SG and CRD. Final approval of the version to be published: RLM, BM, GG, MRL, SG and CRD. Agreement to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved: RLM, BM, GG, MRL, SG and CRD.

Funding This work was supported by unrestricted funds granted to the University of British Columbia from the British Columbia Ministry of Health. (Grant number: not applicable.)

Competing interests All authors have completed the ICMJE uniform disclosure form at www.icmje.org/coi_disclosure.pdf and declare: unrestricted funds granted to the University of British Columbia from the British Columbia Ministry of Health supported this work; BM is acting as an expert witness for Health Canada on a legal case and anticipates future payment for doing so; MRL has received consulting fees from Health Canada and the Conference Board of Canada, and payment for expert testimony from the Attorney General of Canada and the Federation of Post-Secondary Educators; SG has received grants from two government granting

agencies, the Canadian Institutes of Health Research and Alberta Innovates; no other relationships or activities that could appear to have influenced the submitted work

Patient and public involvement Patients and/or the public were involved in the design, or conduct, or reporting, or dissemination plans of this research. Refer to the Methods section for further details.

Patient consent for publication Not applicable.

Ethics approval The study received ethics approval from the University of British Columbia Behavioural Research Ethics Board (H18-03458) and the University of Alberta Health Research Ethics Board (Pro00096201), and all participants provided informed consent. Participants gave informed consent to participate in the study before taking part.

Provenance and peer review Not commissioned; externally peer reviewed.

Data availability statement No data are available.

Supplemental material This content has been supplied by the author(s). It has not been vetted by BMJ Publishing Group Limited (BMJ) and may not have been peer-reviewed. Any opinions or recommendations discussed are solely those of the author(s) and are not endorsed by BMJ. BMJ disclaims all liability and responsibility arising from any reliance placed on the content. Where the content includes any translated material, BMJ does not warrant the accuracy and reliability of the translations (including but not limited to local regulations, clinical guidelines, terminology, drug names and drug dosages), and is not responsible for any error and/or omissions arising from translation and adaptation or otherwise.

Open access This is an open access article distributed in accordance with the Creative Commons Attribution Non Commercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is properly cited, appropriate credit is given, any changes made indicated, and the use is non-commercial. See: http://creativecommons.org/licenses/by-nc/4.0/.

ORCID iDs

Richard L Morrow http://orcid.org/0000-0002-5869-5197 Barbara Mintzes http://orcid.org/0000-0002-8671-915X Scott Garrison http://orcid.org/0000-0002-7024-2158

REFERENCES

- Schmucker C, Schell LK, Portalupi S, et al. Extent of non-publication in cohorts of studies approved by research ethics committees or included in trial registries. *PLoS One* 2014;9:e114023.
- 2 Canestaro WJ, Hendrix N, Bansal A, *et al.* Favorable and publicly funded studies are more likely to be published: a systematic review and meta-analysis. *J Clin Epidemiol* 2017;92:58–68.
- 3 DeVito NJ, Bacon S, Goldacre B. Compliance with legal requirement to report clinical trial results on clinicaltrials.gov: a cohort study. *Lancet* 2020;395:361–9.
- 4 Goldacre B, DeVito NJ, Heneghan C, *et al.* Compliance with requirement to report results on the EU clinical trials register: cohort study and web resource. *BMJ* 2018;362:k3218.
- 5 Alayche M, Cobey KD, Ng JY, *et al.* Evaluating prospective study registration and result reporting of trials conducted in canada from 2009-2019. *Health Policy* [Preprint].
- 6 Kho ME, Brouwers MC. Conference Abstracts of a new oncology drug do not always lead to full publication: proceed with caution. J Clin Epidemiol 2009;62:752–8.
- 7 Lensen S, Jordan V, Showell M, et al. Non-publication and publication bias in reproductive medicine: a cohort analysis. *Hum Reprod* 2017;32:1658–66.
- 8 Sanossian N, Ohanian AG, Saver JL, et al. Frequency and determinants of nonpublication of research in the stroke literature. Stroke 2006;37:2588–92.
- 9 Shamliyan T, Kane RL. Clinical research involving children: registration, completeness, and publication. *Pediatrics* 2012;129:e1291–300.
- 10 Whittington CJ, Kendall T, Fonagy P, et al. Selective serotonin reuptake inhibitors in childhood depression: systematic review of published versus unpublished data. *Lancet* 2004;363:1341–5.
- 11 Chalmers I. Underreporting research is scientific misconduct. JAMA 1990;263:1405–8.
- 12 Chan A-W, Song F, Vickers A, et al. Increasing value and reducing waste: addressing inaccessible research. Lancet 2014;383:257–66.

- 13 Turner EH, Matthews AM, Linardatos E, et al. Selective publication of antidepressant trials and its influence on apparent efficacy. N Engl J Med 2008;358:252–60.
- 14 Chalmers I, Glasziou P, Godlee F. All trials must be registered and the results published. *BMJ* 2013;346:f105.
- 15 Connor JT. Positive reasons for publishing negative findings. *Am J Gastroenterol* 2008;103:2181–3.
- 16 Dowd MD. Breaching the contract: the ethics of nonpublication of research studies. Arch Pediatr Adolesc Med 2004;158:1014–5.
- 17 Goldacre B. *Missing data. bad pharma*. Toronto: McClelland & Stewart, 2012: 1–99.
- 18 Hall R, de Antueno C, Webber A, et al. Publication bias in the medical literature: a review by a Canadian research ethics board. Can J Anaesth 2007;54:380–8.
- 19 Cattapan A, Browne K, Halperin DM, et al. Motivation for participating in phase 1 vaccine trials: comparison of an influenza and an ebola randomized controlled trial. Vaccine 2019;37:289–95.
- 20 Dhalla S, Poole G. Motivators to participation in medical trials: the application of social and personal categorization. *Psychol Health Med* 2013;18:664–75.
- 21 McCann SK, Campbell MK, Entwistle VA. Reasons for participating in randomised controlled trials: conditional altruism and considerations for self [Trials [Electronic Resource]]. *Trials* 2010;11:31.
- 22 Moorcraft SY, Marriott C, Peckitt C, et al. Patients' willingness to participate in clinical trials and their views on aspects of cancer research: results of a prospective patient survey [Trials [Electronic Resource]]. *Trials* 2016;17:.:17.
- 23 Jones CW, Braz VA, McBride SM, et al. Cross-sectional assessment of patient attitudes towards participation in clinical trials: does making results publicly available matter? BMJ Open 2016;6:e013649.
- 24 Morrow RL, Mintzes B, Gray G, et al. Industry sponsor influence in clinical trial reporting in canada: a qualitative interview study. *Clin Ther* 2022;44:374–88.
- 25 Morrow RL, Mintzes B, Gray G, et al. Factors relating to nonpublication and publication bias in clinical trials in Canada: a qualitative interview study. Br J Clin Pharmacol 2023;89:1198–206.
- 26 Guest G, MacQueen K, Namey E. Applied thematic analysis. 2455 Teller Road, Thousand Oaks California 91320 United States: Sage Publications, 2012.
- 27 Creswell JW. Qualitative inquiry & research design: choosing among the five approaches. Thousand Oaks, CA: Sage Publications, 2013.

- 28 Weiss RS. Learning from strangers: the art and method of qualitative interview studies. New York: Free Press, 1994.
- 29 Canadian institutes of health research, natural sciences and engineering research council of canada, and social sciences and humanities research council. Tri-council policy statement: ethical conduct for research involving humans. 2018.
- 30 National Commission for the Protection of Human Subjects of Biomedical and Behavioral Research. The belmont report: ethical principles and guidelines for the protection of human subjects of research. 1979.
- 31 Amstutz A, Schandelmaier S, Frei R, et al. Discontinuation and nonpublication of randomised clinical trials supported by the main public funding body in Switzerland: a retrospective cohort study. BMJ Open 2017;7:e016216.
- 32 Johnson AL, Fladie I, Anderson JM, et al. Rates of discontinuation and nonpublication of head and neck cancer randomized clinical trials. *JAMA Otolaryngol Head Neck Surg* 2020;146:176–82.
- 33 Kasenda B, von Elm E, You J, et al. Prevalence, characteristics, and publication of discontinued randomized trials. JAMA 2014;311:1045–51.
- 34 Tijdink JK, Schipper K, Bouter LM, *et al.* How do scientists perceive the current publication culture? A qualitative focus group interview study among Dutch biomedical researchers. *BMJ Open* 2016;6:e008681.
- 35 ClinicalTrials.gov: united states national library of medicine. n.d. Available: https://clinicaltrials.gov
- 36 Malički M, Marušić A, OPEN (to Overcome failure to Publish nEgative fiNdings) Consortium. Is there a solution to publication bias? researchers call for changes in dissemination of clinical research results. J Clin Epidemiol 2014;67:1103–10.
- 37 Zarin DA, Fain KM, Dobbins HD, *et al.* 10-Year update on study results submitted to ClinicalTrials.gov. *N Engl J Med* 2019;381:1966–74.
- 38 FDA takes action for failure to submit required clinical trial results information to clinicaltrials.gov: united states food and drug administration. 2021. Available: https://www.fda.gov/news-events/ press-announcements/fda-takes-action-failure-submit-requiredclinical-trial-results-information-clinicaltrialsgov
- 39 Rochon PA, Sekeres M, Hoey J, et al. Investigator experiences with financial conflicts of interest in clinical trials. *Trials* 2011;12:9.